Author's response to reviews

Title: Interpersonal psychotherapy (IPT) for late-life depression in general practice: uptake and satisfaction by patients, therapists and physicians.

Authors:

Digna JF van Schaik (a.van.schaik@ggzba.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Aartjan TF Beekman (aartjanb@ggzba.nl)
Marten de Haan (m.dehaan@vumc.nl)
Richard van Dyck (R.van.Dyck@ggzba.nl)

Version: 7 Date: 6 September 2007

Author's response to reviews:

According to the comments in your e-mail of September the 3rd we removed the qualifications and titles (PhD/ MD etc) from the authors names list and did not place each new author on a new line anymore.

We indicated the corresponding author with(*)

We removed the table placement text.

We wrote that the authors declare that they have no competing interests.

We did not place each authors contributions on a new line anymore.

The reference list now contains all authors names.